

W&M ScholarWorks

VIMS Articles

Virginia Institute of Marine Science

7-2020

Challenges in Quantifying Air-Water Carbon Dioxide Flux Using Estuarine Water Quality Data: Case Study for Chesapeake Bay

Marie Herrmann

Raymond G. Najjar

Fei Da Virginia Institute of Marine Science

Marjorie A.M. Friedrichs Virginia Institute of Marine Science

Sreece Goldberger

See next page for additional authors

Follow this and additional works at: https://scholarworks.wm.edu/vimsarticles

Part of the Oceanography Commons

Recommended Citation

Herrmann, Marie; Najjar, Raymond G.; Da, Fei; Friedrichs, Marjorie A.M.; Goldberger, Sreece; Menendez, Alana; Shadwick, Elizabeth; and St-Laurent, Pierre, Challenges in Quantifying Air-Water Carbon Dioxide Flux Using Estuarine Water Quality Data: Case Study for Chesapeake Bay (2020). *Journal of Geophysical Research: Oceans*, 125(7), e2019JC015610. doi: 10.1029/2019JC015610

This Article is brought to you for free and open access by the Virginia Institute of Marine Science at W&M ScholarWorks. It has been accepted for inclusion in VIMS Articles by an authorized administrator of W&M ScholarWorks. For more information, please contact scholarworks@wm.edu.

Authors

Marie Herrmann, Raymond G. Najjar, Fei Da, Marjorie A.M. Friedrichs, Sreece Goldberger, Alana Menendez, Elizabeth Shadwick, and Pierre St-Laurent



JGR Oceans

RESEARCH ARTICLE

10.1029/2019JC015610

Special Section:

Carbon Cycling In Tidal Wetlands And Estuaries Of The Contiguous United States

Key Points:

- The largest source of error in computing the air-water CO₂ flux from historical Chesapeake Bay water quality data is the measurement of pH
- Biological processes are a major cause of the spatial and temporal variability in surface water *p*CO₂ in the mainstem Chesapeake Bay
- Averaged over 1998 to 2018, the mainstem bay was a weak net source of CO₂ to the atmosphere of 1.1–1.4 mol m⁻² yr⁻¹ (95% confidence range)

Supporting Information:

Supporting Information S1

Correspondence to: R. G. Najjar, rgn1@psu.edu

Citation:

Herrmann, M., Najjar, R. G., Da, F., Friedman, J. R., Friedrichs, M. A. M., Goldberger, S., et al. (2020). Challenges in quantifying air-water carbon dioxide flux using estuarine water quality data: Case study for Chesapeake Bay. *Journal* of *Geophysical Research: Oceans*, *125*, e2019JC015610. https://doi.org/ 10.1029/2019JC015610

Received 31 AUG 2019 Accepted 9 MAY 2020 Accepted article online 27 MAY 2020

©2020. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Challenges in Quantifying Air-Water Carbon Dioxide Flux Using Estuarine Water Quality Data: Case Study for Chesapeake Bay

Maria Herrmann¹, Raymond G. Najjar¹, Fei Da², Jaclyn R. Friedman², Marjorie A. M. Friedrichs², Sreece Goldberger¹, Alana Menendez³, Elizabeth H. Shadwick⁴, Edward G. Stets⁵, and Pierre St-Laurent²

¹Department of Meteorology and Atmospheric Science, The Pennsylvania State University, University Park, PA, USA, ²Virginia Institute of Marine Science, College of William & Mary, Gloucester Point, VA, USA, ³Department of Earth and Atmospheric Sciences, The City College of New York, City University of New York, New York, NY, USA, ⁴CSIRO Oceans and Atmosphere, Hobart, Tasmania, Australia, ⁵U.S. Geological Survey, Mounds View, MN, USA

Abstract Estuaries play an uncertain but potentially important role in the global carbon cycle via CO_2 outgassing. The uncertainty mainly stems from the paucity of studies that document the full spatial and temporal variability of estuarine surface water partial pressure of carbon dioxide (pCO_2). Here, we explore the potential of utilizing the abundance of pH data from historical water quality monitoring programs to fill the data void via a case study of the mainstem Chesapeake Bay (eastern United States). We calculate pCO_2 and the air-water CO_2 flux at monthly resolution from 1998 to 2018 from tidal fresh to polyhaline waters, paying special attention to the error estimation. The biggest error is due to the pH measurement error, and errors due to the gas transfer velocity, temporal sampling, the alkalinity mixing model, and the organic alkalinity estimation are 72%, 27%, 15%, and 5%, respectively, of the error due to pH. Seasonal, interannual, and spatial variability in the air-water flux and surface pCO_2 is high, and a correlation analysis with oxygen reveals that this variability is driven largely by biological processes. Averaged over 1998–2018, the mainstem bay is a weak net source of CO_2 to the atmosphere of 1.2 (1.1, 1.4) mol m⁻² yr⁻¹ (best estimate and 95% confidence interval). Our findings suggest that the abundance of historical pH measurements in estuaries around the globe should be mined in order to constrain the large spatial and temporal variability of the CO_2 exchange between estuaries and the atmosphere.

Plain Language Summary Carbon dioxide (CO_2), an important greenhouse gas, is transferred from most estuaries to the atmosphere at a poorly known but potentially large rate. The difficulty in accurately quantifying the transfer is the lack of data of the amount of CO_2 in estuarine surface waters. We evaluated the proposition that the abundance of historical water quality measurements of pH could be used to fill this data gap by conducting an analysis of CO_2 and its transfer to the atmosphere in surface waters of the mainstem of the Chesapeake Bay, a large estuary of eastern United States. The analysis is unusual because of the large number of observations over many years that were used to calculate the CO_2 concentration in estuarine surface waters. CO_2 was found to vary greatly throughout the estuary, from season to season, and from year to year. Using measurements of temperature, salinity, and oxygen, we determined that photosynthesis and respiration had a large influence on surface water CO_2 . Averaged over 1998–2018, the mainstem of the bay released CO_2 to the atmosphere. Our analysis suggests that the approach of using historical water quality measurements should be applied elsewhere to more accurately determine how much CO_2 is exchanged between world estuaries and the atmosphere.

1. Introduction

Estuaries outgas a significant amount of carbon dioxide (CO₂) to the atmosphere, ranging from 0.1 to 0.5 Pg C yr⁻¹, up to one quarter of the uptake of carbon dioxide by the open ocean (Borges & Abril, 2011; Cai, 2011; Chen et al., 2013, 2012; Chen & Borges, 2009; Laruelle et al., 2010). The uncertainty in the outgassing stems from the large spatial and temporal gradients in the partial pressure of CO₂ (pCO₂) in estuarine surface waters, as high as ~100 µatm km⁻¹ and >1,000 µatm month⁻¹, respectively (Borges & Abril, 2011), combined with limited observations. In the most recent global synthesis of estuarine CO₂ outgassing (Chen



et al., 2013), data from 165 estuaries were included, but only 29 systems had seasonal coverage and only 16 of these had published studies of their pCO_2 dynamics (Àlvarez et al., 1999; Frankignoulle et al., 1998; Guo et al., 2009; Jiang et al., 2008; Koné et al., 2009; Mukhopadhyay et al., 2002; Raymond et al., 2000; Zhai et al., 2007). Fewer still had interannual coverage, with 13 of the 16 published seasonal studies covering a time period of less than 2 yr and the remaining 3 consisting of up to nine cruises over as many as 5 yr. Of the studies published since the synthesis of Chen et al. (2013), we are only aware of three that cover more than a few years (Carstensen et al., 2018; Dinauer & Mucci, 2017; Prasad et al., 2013).

This study is based on the premise that historical water quality data can help fill the temporal and spatial variability void in estuarine surface water pCO_2 . Support for this premise is found in the multidecadal pCO_2 studies of Carstensen et al. (2018) and Prasad et al. (2013), who computed pCO_2 using pH and total alkalinity (TA) data from monitoring programs, which increased in number in the 1970s and 1980s in response to a growing public awareness of water pollution. Only recently have monitoring program data sets begun to be used for long-term studies of the estuarine carbonate system, with a focus on acidification and alkalinization (Baumann & Smith, 2017; Carstensen & Duarte, 2019; Duarte et al., 2013, Hu et al., 2015; Najjar et al., 2020; Waldbusser et al., 2013). However, there are at least three challenges associated with using water quality data for studying pCO₂ dynamics. First, water quality monitoring pH and TA data are of relatively low quality when compared to pH and TA measured using modern methods, which translates into a large uncertainty in computed pCO_2 . For example, for conditions typical of a midlatitude estuary (temperature = 15° C, salinity = 15, pH = 8, and TA = 1.5 mol m⁻³), the accuracy of pH probes typically used in water quality monitoring programs (0.2) translates to a pCO_2 accuracy of 50%. Second, monitoring programs either sample on a roughly monthly time interval throughout a system (e.g., Chesapeake Bay Program, 1996), thereby missing variability due to tides, solar diurnal forcing, and weather events or continuously at a few locations within a system (e.g., Baumann & Smith, 2017), thereby missing the large spatial variability from tidal fresh to polyhaline waters. Third, compared to oceanic waters, the carbonate system in estuaries is poorly constrained because of uncertainties about equilibrium constants and organic alkalinity (Dinauer & Mucci, 2017). To our knowledge, no long-term studies of surface water pCO₂ and the air-water CO₂ flux have addressed these multiple challenges and associated sources of error.

In this study, we present an analysis of the surface water pCO_2 and the air-water CO_2 flux for the mainstem Chesapeake Bay using measurements of pH, temperature, and salinity from a water quality monitoring program. The spatial and temporal coverage of the analysis is unprecedented for an estuary, based on monthly or semimonthly surveys of 33 stations throughout the mainstem bay from 1998 to 2018. Special attention is given to error estimation by propagating errors in pH, TA, organic alkalinity, temporal sampling, and the gas transfer velocity. Particularly novel is the use of high-quality measurements of TA and dissolved inorganic carbon (DIC) to better constrain errors. We document seasonal, interannual, and spatial variability of the computed surface water pCO_2 and the air-water CO_2 flux and identify the likely driving factors for this variability through statistical analysis. Dissolved oxygen supersaturation ($\Delta[O_2]$, the negative of the apparent oxygen utilization) is used as a tracer of net ecosystem production (NEP = photosynthesis minus respiration) and a means for assessing NEP's influence on surface water pCO_2 .

2. Study Area, Observations, and Methods

2.1. Study Area

The mainstem Chesapeake Bay is a large, partially stratified, coastal plain estuary located in eastern North America (Figure 1). The standard Chesapeake Bay management segmentation scheme (Chesapeake Bay Program, 2004) partitions the bay into eight segments (Table 1 and Figure 1). Segment 1 is considered tidal fresh, Segment 2 oligohaline, Segments 3–5 mesohaline, and Segments 6–8 polyhaline (see salinity ranges in Table 1). The mainstem Chesapeake Bay ranks high in terms of primary productivity among the world's estuaries (Cloern et al., 2014). Kemp et al. (1997) found the bay to be net heterotrophic in Segments 1–3, presumably resulting from a combination of high turbidity that reduces photosynthesis and high riverine organic carbon loads that stimulate respiration. The clearer waters of Segments 4–5 and 6–8 were found to be metabolically neutral and net autotrophic, respectively. Overall, the bay is thought to be net autotrophic (Kemp et al., 1997, C. Shen et al., 2019), which is unusual among the world's estuaries (Borges & Abril, 2011).





Figure 1. Mainstem bay segmentation and locations of the USGS nontidal alkalinity station and tidal pH, temperature, salinity, and dissolved oxygen stations from the Chesapeake bay Program (CBP). Stations that were used for an evaluation of the pH data quality, as described in the supporting information, are circled (CB5.3 and 5.4). Also shown are Shadwick, Friedrichs, et al. (2019) buoy and Shadwick, De Meo, et al. (2019) stations.

Compared to its condition in the early 1900s, Chesapeake Bay is severely degraded due to eutrophication and the attendant declines in water clarity and dissolved oxygen (see Kemp et al., 2005, for a review). The reduction of water clarity, which may have been exacerbated by overharvesting of oysters and other filter feeders, has historically led to dramatic reductions in underwater grasses, an important benthic habitat. Over the past 30 years, Chesapeake Bay has experienced a modest improvement in over-all water quality, as measured using metrics for chlorophyll a, dissolved oxygen, and water clarity/underwater grasses, consistent with expectations from reductions in riverine nitrogen loading (Zhang et al., 2018).

Studies of the carbonate system of the mainstem Bay are relatively recent. Observations indicate relatively poor acid buffering capacity (Cai et al., 2017); biological sources and sinks of DIC that vary with time, depth, and distance along the axis of the estuary (Brodeur et al., 2019; Friedman et al., 2020; Shadwick, Friedrichs, et al., 2019); and nonconservative TA behavior (Brodeur et al., 2019; Najjar et al., 2020). The only estimates of air-water CO_2 fluxes in the mainstem bay were based on an 8-month data set in mesohaline and polyhaline waters (Friedman et al., 2020), a time series estimate at a single location in the polyhaline bay (Shadwick, Friedrichs, et al., 2019), and numerical models (Shen et al., 2019; St-Laurent et al., 2020). These studies document considerable spatial and temporal variability in the flux, with the mainstem acting as both a source and sink of atmospheric CO_2 , depending on time and location.

2.2. Observations of pH, Temperature, Salinity, and Dissolved Oxygen

Observations of surface water pH, temperature, salinity, and dissolved oxygen are from the Chesapeake Bay Program (CBP), which conducts monthly (semimonthly in the warmer months) bay-wide surveys. We selected 33 stations from the mainstem bay for the analysis (Figure 1 and Table 1), seeking good spatial coverage while avoiding overrepresen-

tation of stations clustered together. Data from the beginning of the regular monitoring program in 1985 through 2018 were downloaded from the CBP's Water Quality Database (http://data.chesapeakebay.net). A close examination of the pH data led us to question some measurements in the early part of the record and limit our analysis to 1998–2018. The data were then subjected to quality control procedures to remove

Characteristics of the Eight Segments of the Mainstern Chesapeake Bay								
Segment	Salinity regime	Seasonal salinity range ^a Area ^b (km ²) Mean depth ^c (m) Station		Station names				
1	Tidal fresh	0-1	152	2.4	CB1.1, CB2.1			
2	Oligohaline	1-6	275	4.5	CB2.2, CB3.1			
3	Mesohaline	5-11	362	6.6	CB3.2, CB3.3C			
4	Mesohaline	9-15	909	10.2	CB4.1C, CB4.2C, CB4.3C, CB4.4			
5	Mesohaline	12-17	1,475	10.5	CB5.1, CB5.1 W, CB5.2, CB5.3, CB5.4, CB5.4 W, CB5.5			
6	Polyhaline	15-20	743	8.7	CB6.1, CB6.2, CB6.3, CB6.4			
7	Polyhaline	18–23	1,521	8.9	CB7.1, CB7.1 N, CB7.1S, CB7.2, CB7.2E, CB7.3,			
					CB7.3E, CB7.4 N, EE3.5			
8	Polyhaline	20-25	412	7.7	CB7.4, CB8.1, CB8.1E			
Bay		13-18	5,849	8.9	33 stations			

Based on the mean annual cycle of the average salinity in each segment, 1998–2018. From Chesapeake Bay Program (2004). Computed as volume (Chesapeake Bay Program, 2004) divided by area.

Table 1





Figure 2. (a) TA predicted from the mixing model as a function of observed TA. (b) Predicted inorganic alkalinity as a function of observed TA. Observed TA is from Shadwick, De Meo, and Friedman (2019).

outliers. Supporting information Text S1 and Figure S1 provide details of the processing of pH, temperature, salinity, and oxygen data.

2.3. Total Alkalinity Mixing Model

TA is measured much less frequently than pH in Chesapeake Bay, and hence, we sought to model surface TA in the bay as a function of measured surface salinity and time. The relationship with salinity recognizes the nearly conservative nature of TA as well as the different TAs of the two main sources of water to the mainstem bay: Atlantic Ocean shelf waters (high-salinity source) and the Susquehanna River (zero-salinity source). The relationship with time recognizes the substantial seasonal and interannual variability of TA of the Susquehanna River. TA in the mainstem bay has increased over the past several decades at a rate that is approximately consistent with expectations from the increase in the riverine source and conservative mixing with a fixed marine endmember (see also Najjar et al., 2020). On the seasonal time scale, however, it is not clear to what extent river TA is felt in the mainstem bay. Hence, we used high-quality measurements of TA from Shadwick, De Meo, and Friedman (2019), which are limited to 2016–2018, to develop and evaluate a TA mixing model made up of two components, one for S < 10, which accounts for seasonal and interannual variability in the riverine endmember, and one for S > 10, which accounts only for the long-term increasing trend in TA in the mainstem bay. Supporting information Text S2 and Figure S2 provide the full details. Results, shown in Figure 2a, reveal a root-mean-square error of 129 mmol m⁻³ for S < 10 (n = 33) and 30 mmol m⁻³ for S > 10 (n = 135).

2.4. Organic Alkalinity Estimation

Organic alkalinity can lead to overestimation of pCO_2 when pCO_2 is computed from pH and TA. Hunt et al. (2011) found overestimation of 13–66% in North American rivers and Abril et al. (2015), in a synthesis of measurements from rivers in Africa, Amazonia, and Europe, found the median overestimation to be 23% but, importantly, showed that the overestimation increases dramatically with decreasing pH, increasing dissolved organic carbon concentration and decreasing TA. For example, the mean overestimation was 15% in rivers with pH > 7 (which have a mean pH of 7.9) and was 362% in rivers with pH < 7 (which have a mean pH of 6.3). Compared to rivers, the overestimation in coastal waters appears to be smaller, ranging from 10 to 47 μ atm (roughly 2% to 10%) in the Baltic Sea (Kuliński et al., 2014), about 1% for typical conditions in Gulf of Mexico coastal waters (Yang et al., 2015), and -7% to 2% in the Scheldt Estuary (Frankignoulle & Borges, 2001).

Unfortunately, there are no readily available estimates of organic alkalinity in Chesapeake Bay. We estimated organic alkalinity by combining CBP pH measurements with simultaneous high-quality measurements of DIC and TA (Shadwick, De Meo, & Friedman, 2019). The inorganic portion of TA was calculated using a MATLAB[®] version of the program CO2SYS (van Heuven et al., 2011) with inputs of pH, temperature, and salinity from the CBP and DIC from Shadwick, De Meo, and Friedman (2019). For this calculation, we used the carbonic acid equilibrium constants of Cai and Wang (1998), the equilibrium



constant for bisulfate ion of Dickson (1990), and the ratio of total boron to salinity of Uppström (1974). Also, the phosphate and silicate concentrations were assumed to be zero as they have a minimal impact on the calculation. We chose the carbonic acid equilibrium constants of Cai and Wang (1998) because, as pointed out by Dinauer and Mucci (2017), these constants agree with the fresh water constants of Millero (1979) whereas other constants for low-salinity waters (Millero, 2010; Millero et al., 2006) do not. The computed inorganic alkalinity is very close to the measured TA (Figure 2b) but is lower, on average, by 20 mmol m⁻³ and has a root-mean-square error of 30 mmol m⁻³. We therefore assumed the organic alkalinity in Chesapeake Bay is normally distributed with mean 20 mmol m⁻³ and ±standard deviation 30 mmol m⁻³. We considered making organic alkalinity a function of salinity but chose not to because the coefficient of determination is small ($r^2 = 0.05$).

Our estimate of organic alkalinity in Chesapeake Bay is within the range of literature values of organic alkalinity in rivers and coastal waters (see review in supporting information Text S5). More specifically, the value of 20 mmol m⁻³ is reasonably close to the value of 10 mmol m⁻³ found by Patsavas et al. (2015) for shelf waters of the eastern United States. An independent estimate of the freshwater endmember can be made using a model of organic alkalinity based on the concentrations of the hydrogen ion and dissolved organic carbon, which was developed from Baltic Sea data (Kuliński et al., 2014). Using a mean hydrogen ion concentration of 0.016 mmol m⁻³ in the tidal fresh region from our analysis and a Susquehanna River dissolved organic carbon concentration of 250 mmol m⁻³ (Fisher et al., 1998; Rochelle-Newall & Fisher, 2002) leads to an organic alkalinity of 23 mmol m⁻³, which is very close to our estimate from DIC, TA, and pH data.

2.5. Calculation of Surface Water pCO₂ and the Air-Water CO₂ Flux

Surface water pCO_2 was computed from the observed surface pH, temperature, and salinity and the modeled TA minus the organic alkalinity (20 mmol m⁻³) at each station at the native temporal resolution of the CBP measurements (roughly monthly, section 2.1) using CO2SYS (van Heuven et al., 2011) with the assumptions described in section 2.4. Figure 3 shows that pCO_2 increases with increasing temperature (*T*), decreasing salinity (*S*), decreasing pH, and increasing TA. Importantly, given the range of conditions in Chesapeake Bay, pCO_2 is roughly an order of magnitude more sensitive to pH than to T, S, and TA. Furthermore, pCO_2 is a highly nonlinear function of pH and a nearly perfect linear function of hydrogen ion concentration ([H⁺]), which becomes important when errors in the pH measurement are propagated in the pCO_2 calculation.

Monthly means of all quantities (pCO_2 , TA, temperature, salinity, and $\Delta[O_2]$) were computed for every station, gaps were filled, averages over each segment were computed, and finally, the air-water CO₂ flux was computed using a standard transfer-velocity formulation (Wanninkhof, 2014) and winds from a reanalysis product. Unless specified otherwise, positive values indicate that the net CO₂ flux is from water to air. All bay-wide averages are computed by weighing segments by area. Full details are provided in supporting information Text S3.

2.6. Error Estimation

We assumed that the main uncertainties in monthly surface water pCO_2 and the air-water CO_2 flux stem from errors in the pH measurement, the TA mixing model, the organic alkalinity estimation, the discrete sampling in time (once or twice per month), and the gas transfer velocity. A Monte Carlo analysis was conducted to propagate these errors to surface water pCO_2 and the air-water CO_2 flux. Other errors, such as those associated with carbonate system equilibrium constants, are not included but are discussed in section 3.1.2.

2.6.1. pH and Alkalinity

For pH, the reported instrument error of the YSI and Hydrolab sondes is $\Delta pH = 0.2$. This error was converted to an error in hydrogen ion concentration using $\Delta[H^+] = 10^{-pH}(10^{-\Delta pH} - 1)$, which is equal to 0.369[H⁺] for $\Delta pH = 0.2$. The TA mixing model error was assumed to be the error in the fit of TA to *S* from Shadwick, De Meo, and Friedman (2019) high-quality data set (Figure 2a, section 2.3). The organic alkalinity error was taken to be $20 \pm 30 \text{ mmol m}^{-3}$, as estimated in section 2.4 and Figure 2b. The errors in pH, the TA mixing model, and organic alkalinity were assumed fixed in time. The error distributions were assumed to be normal in [H⁺], TA, and organic alkalinity. Due to the strong nonlinearity of pCO_2 as a function of pH (Figure 3a), sampling a normal distribution in pH results in a strongly skewed pCO_2 distribution. The





Figure 3. Calculated surface water pCO_2 as a function of (a) *T*, (b) *S*, (c) TA, (d) pH, and (e) [H⁺]. Note a factor of 10 larger vertical axis for pH and [H⁺]. Overlaid are histograms of the surface data. Both pH and [H⁺] are shown to emphasize differences in linearity. When *T*, *S*, TA, and pH (or [H⁺]) are allowed to vary, the other variables are held constant at their approximate mean values for the mainstem Chesapeake Bay surface waters ($T = 18^{\circ}C$, S = 14, TA = 1.6 mol m⁻³, and pH = 8.1).

skewing is positive because the pCO_2 versus pH curve is concave upward. Very little skewing occurs using [H ⁺] and TA because pCO_2 is essentially linear in these variables (Figure 2e). The resulting probability distributions were randomly sampled each time pCO_2 was calculated, to create 1,000 realizations of pCO_2 time series at each station in the bay. Occasionally, an unrealistic value of TA (negative) or [H⁺] (corresponding to pH less than 7 or greater than 9.69, which is outside of the observed pH range at the 33 stations used in the analysis, Figure 3d) was drawn from the distribution, in which case the value was discarded and another draw was made. The resulting time series, which matched the irregular temporal resolution of the Chesapeake Bay monitoring, were then averaged by month.

2.6.2. Discrete Sampling in Time

The monthly averages of pCO_2 at each station are based on one or two pCO_2 computations per month, so are not true monthly averages. To assess the associated sampling error, we used hourly surface pCO_2 estimates at the mouth of the York River Estuary, which is in the polyhaline Chesapeake Bay (Shadwick, Friedrichs, et al., 2019). The pCO_2 estimates were made by using direct measurements of pH, *T*, and *S* and a model of TA as a function of *S*. The data span the time periods December 2016 to March 2017 and January 2017 to May 2018, a total of nine complete months. The time periods (winter and spring) and single location of this data set are not ideal but provide the opportunity to make a first estimate of pCO_2 sampling error in Chesapeake Bay and, as far as we are aware, in any estuary. While Shadwick, Friedrichs, et al. (2019) first computed diurnal averages of pH, *T*, and *S* before computing pCO_2 , we retain the original hourly resolution in order to





Figure 4. Ensemble of instantaneous surface water pCO_2 draws as a function of true monthly mean surface water pCO_2 based on the high-temporal resolution data set of Shadwick, Friedrichs, et al. (2019).

capture diurnal and tidal influences. Our computation also differs from that of Shadwick, Friedrichs, et al. (2019) in that we use the equilibrium constants of Cai and Wang (1998) instead of those of Mehrbach et al. (1973) refit by Dickson and Millero (1987).

To compute the sampling error, we first created a probability distribution at hourly resolution of the times of day that pH was measured in our version of the CBP data set. This probability distribution was then used to randomly sample the hourly pCO_2 data set, assuming no preference on the day of the month. The difference between sampled pCO_2 and the true mean pCO_2 for the month that was sampled was determined to be the sampling error (vertical deviation from the 1:1 line in Figure 4). A sampling error distribution was created by repeating the sampling process 1,000 times. This error distribution has a mean of -9.6μ atm and a standard deviation of 49 μ atm. For each of the 1,000 pCO_2 realizations at the 33 stations in the mainstem Bay (section 2.5), the pCO_2 sampling error was added by drawing from the sampling error distribution. If a draw resulted in a negative pCO_2 , the pCO_2 was set to zero.

2.6.3. Gas Transfer Velocity

Wanninkhof (2014) attempted to quantify the uncertainty in the gas transfer velocity due to the combined influences of uncertainties in the leading coefficient *a*, the Schmidt number, winds, high-wind speed effects, and chemical enhancement. The result was an overall uncertainty of 20%, which is applicable to global-scale estimates of air-sea CO_2 exchange.

This is not a rigorous estimate, and it is unclear if 20% is one standard error or half the 95% confidence range. Quantifying the error at the local scale in a rigorous manner is not possible at this time. To be conservative, we used a standard error estimate of 30%. Specifically, every time the gas transfer velocity was computed, which is once per month per segment, we used a Monte Carlo approach and randomly added an error chosen from a normal distribution with a mean of 0 and standard deviation of 30%.

2.6.4. Error Reporting Conventions

We use the following convention to report the pCO_2 and CO_2 flux calculation results. The results of a single pCO_2 and CO_2 flux calculation (i.e., without random errors in pH, the alkalinity mixing model, organic alkalinity, temporal sampling, and the gas transfer velocity) are referred to as *point estimates*. The results of Monte Carlo calculations are reported as *best estimates* (means) with 95% *confidence intervals* (2.5th and 97.5th percentiles) or standard errors (standard deviations) of each corresponding ensemble. Unless otherwise stated, all results reported without uncertainties are point estimates, and those reported with a confidence range or standard error are best estimates.

3. Results

3.1. *p*CO₂ Variability

3.1.1. Temporal Variability

Surface water pCO_2 is highly variable in time, as time series (Figure 5) and basic statistics (Table 2) for each segment reveal. Segments 1 and 2 (tidal fresh and oligohaline regions) stand out as particularly variable in surface water pCO_2 , with the highest coefficients of variation, 66% and 65%, respectively, compared to 34–57% in the rest of the bay. Similarly, Segments 1 and 2 have the most prominent annual cycles, making up 56% and 69%, respectively, of the variability of the full time series, compared to 14–33% in the mesohaline and northern polyhaline bay (segments 3–6) and <5% in the southern polyhaline bay (Segments 7–8).

Figure 6 shows that mean annual cycle is well defined in Segments 1 and 2 (tidal fresh and oligohaline region) and Segments 4–7 (central mesohaline to central polyhaline region), with the saltier segments lagging behind the fresher segments by 0–3 months. From the fresher segments to the saltier segments, minima shift from winter to typically spring and maxima shift less, from summer to sometimes early fall (Table 2).

Biological control on the temporal variability in surface water pCO_2 is suggested by the anticorrelation between pCO_2 and $\Delta[O_2]$, which is clearly seen in mean annual cycles of these two quantities (Figure 6).





Figure 5. Temporal variability of the best estimates of the surface water pCO_2 computed for eight segments (Figure 1) of the mainstem Chesapeake Bay.

The extrema in the two annual cycles line up well. The lag of the lower bay with respect to the upper bay that is seen in pCO_2 is also seen in $\Delta[O_2]$. Table 3 shows that $\Delta[O_2]$ accounts for 47% to 90% of the variability in the annual cycle of pCO_2 in six of the eight segments. Table 3 also shows >95% significance in the relationship between the two variables for the full time series in all segments, with 25% to 43% of the variability accounted for in Segments 1–5. Even with the mean annual cycle removed, all segments show correlations of greater than 95% confidence between the two variables. Only for annual averages are correlations generally poor.

Table 2

Statistics of the Best Estimates of Monthly Average Surface Water pCO_2 in Each Segment and the Bay-Wide Average							
	Mean + 1 standard	Coefficient	Mean annual cycle				
Segment	deviation (µatm)	of variation	Variance fraction	ariance fraction Month of minimum			
1	803 ± 534	0.66	0.56	February	September		
2	1,139 ± 740	0.65	0.69	February	July		
3	576 ± 309	0.54	0.22	March	September		
4	378 ± 192	0.51	0.28	March	October		
5	345 ± 145	0.42	0.33	May	September		
6	373 ± 214	0.57	0.14	May	October		
7	449 ± 196	0.44	0.04	May	September		
8	527 ± 244	0.46	0.01	January	August		
Bay	457 ± 154	0.34	0.25	March	September		

Streamflow has the potential to profoundly impact estuarine pCO_2 through its influence on salinity, stratification, and the delivery of inorganic nutrients, organic matter, exogenous DIC, and sediment. To investigate the potential influence of streamflow on pCO_2 , correlation statistics are presented with salinity (Table 3). We prefer using salinity instead of streamflow in the correlation analysis because it takes time, sometimes several months, for a change in flow and other characteristics of the Susquehanna River to be felt



Figure 6. Mean annual cycles of the best estimates of surface water pCO_2 and oxygen supersaturation ($\Delta[O_2]$) in eight segments (Figure 1) of the mainstem Chesapeake Bay. Note the change in vertical axes among the panels. Dashed lines show atmospheric pCO_2 (blue) and oxygen saturation (red). Gray shading shows 95% confidence range on pCO_2 .



Table 3

Coefficient of Determination (r^2) of Surface Water pCO₂ (Best Estimates) Versus Oxygen Supersaturation and Salinity for Different Time Frames

Segment	Oxygen supersaturation				Salinity			
	Full time series	Annual averages	Anomalies	Mean annual cycle	Full time series	Annual averages	Anomalies	Mean annual cycle
1	0.25	0.33	0.18	0.47	0.11	0.36	0.08	
2	0.35	0.21	0.14	0.81	0.05	0.45	0.06	
3	0.37	0.23	0.42					
4	0.43		0.35	0.74	0.15	0.21	0.06	0.64
5	0.31		0.16	0.80	0.39	0.69	0.28	0.75
6	0.09		0.03	0.90	0.20	0.22	0.11	0.87
7	0.03		0.02	0.55	0.20	0.32	0.20	0.47
8	0.04		0.03		0.10		0.12	
Bay	0.18		0.09	0.66	0.25	0.48	0.23	

Note. Anomalies are the full time series with the mean annual cycle removed. Values are shown when there is >95% confidence that the slope of the relationship is different from zero; under these conditions, all correlation coefficients (*r*) are negative with oxygen supersaturation and positive with salinity.

throughout the mainstem bay (Shen & Wang, 2007). There is a close correspondence between pCO_2 and salinity for the four time scales considered in Table 3. All of the significant correlations are positive. Correlations are particularly strong for annual averages, where 21–69% of the variability in pCO_2 is accounted for by salinity in six of the eight segments. Correlations are also strong for the annual cycle in Segments 4–7, where salinity accounts for 47–87% of the variability in pCO_2 . A positive correlation between salinity and pCO_2 for constant temperature, DIC, and TA is expected through the salinity dependence of the equilibrium constants, but the sensitivity found here is higher. For example, for the bay-wide average correlations (last row in Table 3), the slopes are equivalent to 6–7% per salinity unit, while the expectation from the carbonate equilibria is 4% per salinity unit with *T*, *S*, and DIC and TA at their bay-wide average values of 15.4°C, 15.3, 1,462 mmol m⁻³, and 1,537 mmol m⁻³, respectively. The positive pCO_2 -salinity correlation indicates that periods of high flow are associated with lower pCO_2 . In a net autotrophic system like Chesapeake Bay (Kemp et al., 1997), this effect may have more of an influence than the simultaneous delivery of sediment and organic matter, which would tend to push the estuary toward heterotrophy via light limitation by sediment and enhanced respiration due to organic matter.

3.1.2. Spatial Variability

Spatial variability in pCO_2 is considerable, with the best estimates for the long-term averages varying among segments by a factor of 3 (Table 2 and Figure 7): from a low of 345 (336, 352) μ atm in Segment 5 (mesohaline)



Figure 7. Long-term averages of the best estimates of surface water pCO_2 and oxygen supersaturation ($\Delta[O_2]$) in eight segments (Figure 1) of the mainstem Chesapeake Bay as a function of latitude. Dashed lines show atmospheric pCO_2 (blue) and oxygen saturation (red). Error bars show 95% confidence range on pCO_2 .

to a high of 1,139 (1,098, 1,177) μ atm in Segment 2 (oligohaline). Similar to the temporal variability of *p*CO₂, the spatial variability of *p*CO₂ appears to be biologically driven, as there is a strong anticorrelation with Δ [O₂] across the eight segments ($r^2 = 0.91$, p < 0.001, Figure 7).

3.1.3. Degree of Supersaturation

Figures 5–7 show that bay surface waters vary in their degree of saturation with respect to atmospheric CO_2 . The bay-wide, long-term average surface water pCO_2 is 457 (452, 462) µatm, which is 17% greater than the corresponding atmospheric value, 381 µatm. Segments vary from being, on average, highly supersaturated (199% in Segment 2) to slightly undersaturated (-10%, Segment 5). Undersaturation occurs in all segments, but the fraction of time this occurs varies greatly, as seen in Figure 5, from only 9% in Segment 2 to 66% in Segment 6. The month of most frequent undersaturation by segment is always in winter or spring (January–June). As can be seen in Figures 5 and 6, the seasonality of the degree of saturation



is almost completely controlled by surface water pCO_2 , which has a range (maximum minus minimum) in its mean annual cycle that varies from 97 µatm in Segment 8 to 1,628 µatm in Segment 2, much greater than the range in the mean annual cycle of atmospheric pCO_2 (15–17 µatm among the eight segments).

3.2. Air-Water Flux

The air-water CO_2 flux is the product of the pCO_2 gradient across the air-water interface, the CO_2 gas transfer velocity, and the CO_2 gas solubility (Equation S3). We have already noted that variability in the pCO_2 gradient is driven almost completely by surface water pCO_2 (section 3.1.3). Temperature has a large impact on CO_2 gas solubility (α) and, through the Schmidt number, on the gas transfer velocity (k_w). However, temperature effects on the product $k_w \alpha$ nearly cancel, and so it is convenient to define this product as the gas transfer coefficient, k_g , whose variability is mainly due to wind speed (Etcheto & Merlivat, 1988), with some minor variability due to salinity and total air pressure.

In the mainstem bay, k_g has considerable temporal (Figures S3a and S3b) and spatial (Figure S3c) variability, both of which, as expected from the formulation of the gas transfer velocity adopted here (Equation S4), are dominated by the variability in the square of the wind speed. Temporal variability in the bay-averaged k_g is largely seasonal, with the mean annual cycle accounting for 66% of the variability of the full time series and characterized by a maximum in January that is 2.5 times higher than the minimum, which occurs in July (Figure S3b). There is also a modest but significant (p = 0.02) declining trend in the bay-averaged k_g anomaly of -9.04×10^{-6} mol m⁻² month⁻¹ µatm⁻¹ yr⁻¹, which corresponds to a decline of 11% from 1998 to 2018 (given the mean k_g of 0.00174 mol m⁻² month⁻¹ µatm⁻¹). Temporal variability in individual segments is similar to the bay-wide average, with strong annual cycles that account for 58% to 76% of the variability of the full time series. Spatial variability of long-term averages shows a slight decline from Segments 1 to 4, an increase of 47% from Segments 4 to 5, and a modest decline from Segments 5 to 8.

Averaged over the bay, the air-water CO_2 flux is highly variable, with a prominent annual cycle, and fewer periods of CO_2 outgassing toward the end of the study period (Figure 8a). To a large extent, the variability of the bay-averaged air-water CO_2 flux mimics the variability of the bay-averaged air-water pCO_2 gradient, with both having strong mean annual cycles with spring minima and fall maxima (Table 2 and Figure 8b). Spatial variability in the CO_2 flux, with high CO_2 outgassing in the upper bay and weak-tomoderate CO_2 outgassing in the lower bay (Figure 8c), is also largely driven by spatial variability in the air-water pCO_2 gradient (Figure 7).

On average the bay outgasses CO₂ at rate of 1.2 (1.1, 1.4) mol m⁻² yr⁻¹. The mean flux is only slightly less than the product of the mean gas transfer coefficient (0.0209 mol m⁻² yr⁻¹ μ atm⁻¹) and the mean air-water *p*CO₂ gradient (66 μ atm), which is equal to 1.4 mol m⁻² yr⁻¹. Thus, covariations between the gas transfer coefficient and the air-water *p*CO₂ gradient are modest, due to the fact that most of the variability in k_g is seasonal and 90° out of phase with the bay-wide average air-water *p*CO₂ gradient (e.g., the maximum in k_g is in winter and the maximum of the gradient is in fall). In fact, the product of the mean k_g and mean air-water *p*CO₂ gradient departs substantially from the mean flux only in Segments 1 and 2 (Figure 8c). In these segments, this product exceeds the mean flux because the mean flux accounts for the negative covariation between k_g and the air-water *p*CO₂ gradient, which results from the mean annual cycles of these two quantities being nearly 180° out of phase (e.g., compare the air-water *p*CO₂ gradient in Figure 6a with k_g in Figure S3b).

The air-water CO₂ flux integrated over each segment has a substantially different pattern with latitude than the flux per unit area (Figure 8c) because of the large range in the areas of the eight segments (Table 1); for example, the largest segment (7) is 10 times the area of the smallest segment (1). The bay-wide integrated flux is 84.9 Gg yr⁻¹, with Segments 1 + 2 + 3 outgassing 58.8 Gg yr⁻¹, Segments 4 + 5 ingassing 16.0 Gg yr⁻¹, and Segments 7 + 8 outgassing 41.1 Gg yr⁻¹.

The bay-averaged annual air-water CO₂ fluxes vary considerably over the study period, from -0.8 (-1.3, -0.4) mol m⁻² yr⁻¹ in 2006 to 7.2 (6.1, 8.1) mol m⁻² yr⁻¹ in 1999 (Figure 8a). Closer examination of the annual fluxes reveals that none of the annual fluxes during the first 5 years (1998–2002) were exceeded in the following years. These unusually high annual fluxes, with an average (± 1 standard deviation) of 4.1 \pm 1.9 mol m⁻² yr⁻¹, are a stark contrast to the rest of the record, which has an average flux of





Figure 8. Best estimates of the net flux (with 95% confidence intervals) of CO_2 from water to air for Chesapeake Bay. Bay-wide average (a) monthly (blue) and annual (red) time series and (b) mean annual cycle. (c) Long-term averages in flux per unit area (blue) and integrated flux (red). The blue x's are the mean gas transfer coefficient multiplied by the mean pCO_2 gradient across the air-water interface.

 $0.3 \pm 1.0 \text{ mol m}^{-2} \text{ yr}^{-1}$. These high fluxes are related to relative high salinity (16.6 ± 1.9) compared to the rest of the record (15.3 ± 1.7), which is due to the relatively low streamflow (Figure S2a). As discussed in section 3.1.1, a net autotrophic system such as Chesapeake Bay might be expected to be pushed toward more heterotrophy (and hence higher CO₂ outgassing) under low-flow years.

3.3. Uncertainty Analysis

We accounted for errors in the pH measurement, the alkalinity mixing model, the organic alkalinity estimation, temporal sampling, and the gas transfer velocity and now compare them for the calculation of the longterm, bay-wide averages of the surface water pCO₂ and air-water CO₂ flux (Figure 9). Six cases are considered: all errors and each of the five errors alone. The best estimate is always slightly higher than the point estimate as a result of weak nonlinearities in the surface water pCO_2 calculation. The biggest error in the surface water pCO_2 calculation is the pH measurement error (Figure 9a). The same is true of the air-water CO_2 flux calculation, with errors, the gas transfer velocity, temporal sampling, the alkalinity mixing model, and the organic alkalinity estimation having 95% confidence ranges in surface water pCO_2 that are 72%, 27%, 15%, and 5% of the same range due to the pH measurement error (Figure 9b). Errors may be large for an individual month for an individual segment, but they decrease dramatically when averaged over time and space. Figure 10a shows an example that starts with a fairly large error (95% confidence range) in the surface pCO_2 for November 2009 in Segment 7 (290 µatm) and gets progressively smaller for the 2009 average in that segment (117 µatm), the long-term average in that Segment (20 µatm), and then finally for the long-term, bay-wide average (10 µatm). Similar reductions are seen for the air-water CO₂ flux (Figure 10b). The dramatic error reduction due to averaging emphasizes one of the advantages of water quality monitoring in Chesapeake Bay, which is characterized by a long and regular record with excellent spatial and temporal coverage.





Figure 9. Best estimate and 95% confidence ranges with the point estimate removed for the long-term, bay-wide averages of the (a) surface water pCO_2 and (b) air-water CO_2 flux. Six cases are considered: all errors, the pH measurement error, the alkalinity mixing model error, the organic alkalinity estimation error, the temporal sampling error, and the gas transfer velocity error.



Figure 10. Example of the effect of temporal and spatial averaging on the error (95% confidence range) in (a) surface water pCO_2 and (b) the air-water CO_2 flux. The bars in each panel represent, in order, segment 7 November 2009 monthly average, segment 7 2009 annual average, segment 7 long-term average, and the bay-wide long-term average.



4. Discussion

4.1. Other Uncertainties

4.1.1. Uncertainties in Carbonate System Equilibria

In addition to uncertainties in pH, the TA mixing model, organic alkalinity, temporal sampling, and the gas transfer velocity, there are uncertainties in the equations of the carbonate system equilibria, which are used to compute pCO_2 from pH and TA. Such uncertainties stem from the assumption of zero phosphate and silicate concentration and uncertainties in carbonate system equilibrium constants. Each of these is considered now.

Phosphate and silicate concentrations in surface waters of Chesapeake Bay can be as high as 1.5 and 100 μ mol kg⁻¹, respectively (Malone et al., 1996). For the mean conditions described above (Figure 3, section 2.5), changing phosphate and silicate concentrations to these concentration decreases surface water pCO_2 by only 0.4 and 0.6 μ atm, respectively. Hence, the assumption of zero concentration for these nutrients is reasonable.

Random errors in the carbonic acid equilibrium constants also translate to small errors in pCO_2 , about 2 µatm, when computed from pH and TA (Emerson & Hedges, 2008, page 115). However, systematic errors may be substantial, particularly in low-salinity waters (Dinauer & Mucci, 2017). In addition to the carbonic acid constants of Cai and Wang (1998), which were used here and are applicable for S = 0-40, the CO2SYS program (van Heuven et al., 2011) has two other sets of equilibrium constants applicable for estuarine waters: Millero (2010) for S = 0-40 and Millero et al. (2006) for S = 1-50. We compared best estimates of the air-water CO_2 flux and surface water pCO_2 computed with these three sets of constants as well as the constants of Mehrbach et al. (1973) as refit by Dickson and Millero (1987), which are applicable for 20 < S < 40, are widely believed to be the most accurate for oceanic waters (e.g., Emerson & Hedges, 2008), and gave good results when comparing pCO₂ computed from pH and TA with measured pCO2 in estuarine conditions (Frankignoulle & Borges, 2001). The bay-wide, long-term average CO2 outgassing using the constants of Millero (2010), Millero et al. (2006), and Mehrbach et al. (1973) was 47%, 46%, and 23% higher, respectively, than that using the constants of Cai and Wang (1998). Surface water pCO_2 computed using the constants of Millero (2010) was higher than that computed using the constants of Cai and Wang (1998) by an average of 305 µatm in Segment 1 and decreasing dramatically and monotonically to 5 µatm in Segment 8, consistent with the findings of Dinauer and Mucci (2017). Differences between the constants of Millero et al. (2006) and Millero (2010) were everywhere small, with average differences in surface water pCO_2 of 0–2 µatm in all segments. Surface water pCO_2 computed with the constants of Mehrbach et al. (1973) was lower in Segments 1 and 2 (114 and 20 µatm, respectively) and slightly higher in Segments 3–8 (9–23 µatm) than that computed with the constants of Cai and Wang (1998).

4.1.2. Additional Uncertainties in the Gas Transfer Velocity

The gas transfer velocity parameterization used here (Wanninkhof, 2014) was designed primarily for use in the open ocean. The main difference in the gas transfer velocity between estuaries and the open ocean is that a significant fraction of the turbulence that regulates gas exchange in estuaries may originate from shear at the sea floor resulting from tidal currents (Ho et al., 2016). However, we show in supporting information Text S4 that it is unlikely that tidal currents have a significant impact on the gas transfer velocity in Chesapeake Bay, which is relatively deep and has relatively weak tidal currents. There are other gas transfer processes distinct to estuaries, including fetch and turbidity. However, the impact of these factors is highly uncertain because parameterizations of these processes are largely untested (Borges & Abril, 2011).

4.2. Comparison With Other Estuarine Air-Water CO₂ Flux Studies

We compared our air-water CO_2 flux results with three recent studies in the mainstem bay. Shen, Testa, Li, et al. (2019) employed a biogeochemical model to estimate the air-water CO_2 flux in the mainstem bay for the year 2016 and found a net uptake of 2.2 mol C m⁻² yr⁻¹ or 130 Gg C yr⁻¹ over an area of 5,050 km², somewhat smaller than the area used here. Shen, Testa, Ni, et al. (2019) extended the model simulation to the 1985–2016 period and found that the average ± 1 standard deviation of the annual mainstem bay fluxes was -39 ± 112 Gg C yr⁻¹, indicating uptake, on average, but with large interannual variability. Friedman et al. (2020) estimated the air-water CO_2 flux from surface water observations of pCO_2 during four cruises from fall 2016 to summer 2017 in the mesohaline and polyhaline bay to find a net uptake of 0.38 mol m

 $^{-2}$ yr⁻¹, which translates to 23 Gg C yr⁻¹ using the sum of areas 4–8 in Table 1. Shen, Testa, Ni, et al. (2019) found relatively high rates of CO₂ outgassing in the upper bay (Segments 1–3) during 2016: 8.8 mol C m⁻² yr⁻¹, which agrees well with our estimate of 10.0 (7.5, 12.8) mol C m⁻² yr⁻¹ for the same region and year. In the middle bay (Segments 4 and 5), Shen, Testa, Li, et al. (2019) and Friedman et al. (2020) found an uptake of approximately 3 mol C m⁻² yr⁻¹ during 2016 and during a time period from fall 2016 to summer 2017, respectively. These estimates contrast with our estimate of a weak CO₂ outgassing at a rate of 0.7 (0.0, 1.5) mol C m⁻² yr⁻¹ in 2016 and near-neutral flux of -0.2 (-0.9, 0.6) mol C m⁻² yr⁻¹ from fall 2016 through summer 2017 (negative flux indicates uptake). In the lower bay (segments 6–8) in 2016 we estimated near neutral condition of–0.1 (-0.8, 0.6) mol C m⁻² yr⁻¹, which is in agreement with the findings of Shen, Testa, Li, et al. (2019) who also found neutral conditions in 2016. In the same geographic region but during a period from fall 2016 to summer 2017, Friedman et al. (2020) found and uptake of about 0.5 mol C m⁻² yr⁻¹, which is slightly outside our flux range of between -0.3 and 1.1 mol C m⁻² yr⁻¹.

Although Chesapeake Bay is typical of other estuaries in that it is a source of CO₂ to the atmosphere, it is a very weak source. In the global synthesis of Chen et al. (2013), the mean area-weighted estuarine CO₂ outgassing rate is 7.7 mol C m⁻² yr⁻¹, a factor of 7 greater than the mean outgassing of Chesapeake Bay found in our study (1.1–1.4 mol C m⁻² yr⁻¹). The bay is also is a weak source compared to estuaries of eastern North America, which outgas CO₂ at a rate of 9 ± 4 (mean ± 2 standard errors) mol C m⁻² yr⁻¹. Chen et al. (2013) found the mean CO₂ outgassing for S < 2 to be 39.0 mol C m⁻² yr⁻¹ and for 2 > S > 25 to be 17.5 mol C m $^{-2}$ yr⁻¹, which can be compared to Chesapeake Bay's mean outgassing rates for Segments 1–2 and 3–8: 8.6 and 0.5 mol C m⁻² yr⁻¹, respectively. Hence, Chesapeake Bay follows the global outgassing pattern with salinity, albeit at much lower rates. The relative weakness of Chesapeake Bay's outgassing is consistent with it being net autotrophic, while most of the world's estuaries are heterotrophic (Borges & Abril, 2011). Note that, due to lateral advection, a net autotrophic system need not be a sink of atmospheric CO₂.

4.3. Inorganic Carbon Budget of the Mainstem Chesapeake Bay

The results of this study in combination with studies on the riverine input of DIC and NEP allow an inorganic carbon budget to be developed for the mainstem bay. If averages over a year or more are considered, then it is reasonable to assume that the mainstem bay is in steady state, with riverine input (R) and lateral transport (L) of DIC balanced by losses due to NEP and air-water exchange of CO₂ (f):

$$R + L = NEP + f \tag{1}$$

where the dimensions of the equation are mass per unit time. Here we have assumed that sources and sinks of calcium carbonate are relatively small, which was suggested by the near-neutral alkalinity budget of the mainstem bay developed by Brodeur et al. (2019). Lateral transport represents advection and turbulent diffusion of DIC at the boundaries of the mainstem bay with Atlantic Ocean waters at the mouth of the bay, major tributaries of the mainstem bay (e.g., the Potomac, James, York, and Rappahannock River estuaries), and fringing marshes. Lateral transport is difficult to measure and is computed here as a residual term of Equation 1 after estimates of R, NEP, and f are made.

Riverine input of DIC to the Chesapeake Bay mainstem from the Susquehanna River was estimated by Stets and Striegl (2012) to be 0.27 Tg C yr⁻¹. We update this flux here using the approach for generating the river alkalinity time series (supporting information Text S2). A time series of daily DIC fluxes in the Susquehanna River at Conowingo, Maryland (USGS gauge number 01578310; USGS, 2019) was constructed for 1 October 1967 to 30 September 2016 by first computing DIC concentrations from pH and TA, as in Stets and Striegl (2012), and then applying a statistical model (Hirsch et al., 2010); effective monthly average DIC concentrations were computed as the monthly fluxes divided by the monthly streamflow. To gap fill effective monthly average concentrations for October 2016 to December 2018, DIC fluxes at monthly resolution for the period of interest here, 1998–2018, were calculated as a product of effective monthly concentrations and monthly flows and then averaged annually. The mean annual flux (± 1 standard deviation, taken here to represent the uncertainty on the long-term annual mean) is equal to 0.42 \pm 0.13 Tg C yr⁻¹. This is higher than the flux estimated by Stets and Striegl (2012) because of the different time period covered by that study.



There are several estimates of NEP in the mainstem bay. Kemp et al. (1997) employed five independent methods based on incubations and the budgets of carbon, nitrogen, and phosphorus, finding the mean of these approaches (\pm its standard error) to be 4.2 ± 0.6 mol C m⁻² yr⁻¹ or 0.29 ± 0.04 Tg C yr⁻¹ using the area in Table 1. Much of the data that went into these estimates are from the late 1970s to early 1990s. Feng et al. (2015) applied a nitrogen-based biogeochemical model over the period 2001–2005 to find annual NEP (\pm 1 standard deviation of the five annual averages) to be $74 \pm 23 \times 10^9$ g N yr⁻¹, which translates to 0.46 ± 0.14 Tg C yr⁻¹ using a C:N ratio of 106:17 (Hedges et al., 2002). Finally, Friedman et al. (2020) estimated NEP to be -0.5 mol C m⁻² yr⁻¹ by constructing a DIC budget in the mesohaline and polyhaline mainstem bay from fall 2016 to the summer of 2017. The range in these estimates is not surprising given the different approaches, time periods, and spatial extent of the studies. Given the limited spatial and temporal extent of the Friedman et al. (2020) estimate, we do not consider it further here. Rather, we take the average of the estimates of Kemp et al. (1997) and Feng et al. (2015) to arrive at a mean NEP (\pm 1 standard error) of 0.38 ± 0.15 Tg C yr⁻¹, where we assumed that the standard deviation of the Feng et al. (2015) estimate represents the standard error and that the errors from the two studies are independent.

Combining our estimated net flux of CO_2 from the mainstem bay to the atmosphere of 0.085 ± 0.005 Tg C yr⁻¹ with the estimates of riverine input and NEP described above and Equation 1 leads to a net lateral input of DIC to the mainstem bay of 0.04 ± 0.20 Tg C yr⁻¹. We have computed the uncertainty in *L* by propagating the above uncertainties in *R*, NEP, and *f*, assuming the uncertainties to be independent of each other. The positive value of *L* indicates that there is likely a source, albeit weak, of DIC to the mainstem bay from lateral advection. This source, for example, could be due to the residual circulation at the mouth of the bay, which transports DIC-rich water into the bay at depth and DIC-poor water out of the bay near the surface. Our findings can be compared with those of Brodeur et al. (2019), who applied a 1-D, steady-state mixing model to 12 monthly surveys of the mainstem bay and found that the bay exports DIC at rate of 0.29 ± 0.06 Tg C yr⁻¹ at its mouth. Our findings are not necessarily inconsistent with those of Brodeur et al. (2019) because of the different time periods considered as well as the other locations where DIC can enter the mainstem bay (e.g., from the tributaries and fringing marshes).

A local view of the carbon balance is afforded by comparisons of NEP with the air-water CO₂ flux in three main portions of the bay that correspond fairly closely to those used by Kemp et al. (1997) in their NEP analysis based on incubations: the upper, middle, and lower bay, which encompass Segments 1–3, 4–5, and 6–8, respectively (Figure 1). In the upper bay, Kemp et al. (1997) found NEP to be $-7.3 \text{ mol C m}^{-2} \text{ yr}^{-1}$, indicating net heterotrophic conditions or net production of CO₂. For our study period, we find that the mean net CO₂ flux from the estuary to the atmosphere in the upper bay is 6.2 (5.8, 6.6) mol C m⁻² yr⁻¹, which nearly balances Kemp et al. 's (1997) NEP in the upper bay, indicating that net lateral fluxes of DIC into or out of this region are small. In the middle bay, there is rough agreement between Kemp et al. 's (1997) NEP of 0.0 mol C m⁻² yr⁻¹ and our weak CO₂ flux of $-0.5 (-0.7, -0.3) \text{ mol C m}^{-2} \text{ yr}^{-1}$, both of which indicate near-neutral conditions. The lower bay is where the two fluxes diverge the most, with Kemp et al. 's (1997) NEP equal to 7.7 mol C m⁻² yr⁻¹ and our CO₂ outgassing of 1.3 (1.1, 1.5) mol C m⁻² yr⁻¹, which implies, at steady state, a net lateral transport into the lower bay of about 9 mol C m⁻² yr⁻¹.

In summary, we find that the air-water CO_2 flux is a modest component of the inorganic carbon budget for the mainstem bay as a whole but can be locally significant. CO_2 outgassing accounts for only about a fifth of the net DIC sinks in the mainstem bay (CO_2 outgassing + NEP). Locally, the CO_2 outgassing flux may either be the main way DIC produced from NEP is removed from the system (e.g., in the upper bay) or a small fraction of the net lateral influx (e.g., in the lower bay).

5. Summary and Conclusions

An analysis of surface water pCO_2 and the air-water CO_2 flux has been conducted for eight segments of the mainstem Chesapeake Bay at monthly resolution for a 21-year period (1998–2018) by exploiting pH, TA, temperature, and salinity measurements from a water quality monitoring program. To our knowledge, this is the first attempt to simultaneously propagate errors in multiple factors, including pH, TA, organic alkalinity, temporal sampling, and the gas transfer velocity. The main findings are as follows:



- 1. The largest error results from the measurement of surface water pH, closely followed by the error in the gas transfer velocity. Errors in the alkalinity mixing model and temporal sampling were secondary, but substantial, while the error in organic alkalinity was estimated to be very small.
- 2. On average the bay is a weak source of CO_2 to the atmosphere, with a net flux of 1.2 (1.1, 1.4) mol m⁻² yr ⁻¹ or 85 (75, 95) Gg yr⁻¹.
- 3. Surface water pCO_2 is highly variable in time, particularly in the tidal fresh and oligohaline bay, where there is a coefficient of variation as high as 66% and a fraction of the variability accounted for by the mean annual cycle as high as 69%.
- 4. There is a shift in the annual cycle of surface water pCO_2 from the fresher segments to the saltier segments of the bay, with minima shifting from winter to typically spring and maxima shifting less, from summer to sometimes early fall.
- 5. Biological control on the seasonal, interannual, and spatial variability in surface water pCO_2 is indicated by the strong anticorrelation between pCO_2 and oxygen supersaturation.
- 6. pCO_2 and salinity are positively and significantly correlated over seasonal and interannual timescales, even after correcting for the salinity dependence of the carbonate system equilibrium constants, suggesting that periods of high streamflow lead to more autotrophy.

Although historic pH and alkalinity measurements from water quality monitoring programs are generally of lower quality than those now made routinely by the marine chemistry research community, the shear abundance of them allows random measurement errors to be reduced to the point where meaningful insights can be gained about spatial and temporal variability in important derived quantities, such as surface water pCO_2 and the air-water CO_2 flux. We demonstrated that high-quality and high-resolution carbonate system measurements of limited spatial and temporal coverage can be paired successfully with the abundant historical measurements of pH and salinity to facilitate bay-wide analysis of carbonate system over multiple decades. Our findings suggest that the abundance of historical pH measurements in estuaries around the globe should be mined in order to constrain the large spatial and temporal variability of the CO_2 exchange between estuaries and the atmosphere, leading to an improved understanding of the role of estuaries in the global carbon cycle.

Data Availability Statement

Data used in this study are archived at The Pennsylvania State University's institutional repository, ScholarSphere (https://doi.org/10.26207/bm5n-n185).

References

- Abril, G., Bouillon, S., Darchambeau, F., Teodoru, C. R., Marwick, T. R., Tamooh, F., et al. (2015). Large overestimation of pCO₂ calculated from pH and alkalinity in acidic, organic-rich freshwaters. *Biogeosciences*, *12*(1), 67–78. https://doi.org/10.5194/bg-12-67-2015
 Àlvarez, M., Fernández, E., & Pérez, F. F. (1999). Air–sea CO₂ fluxes in a coastal embayment affected by upwelling: Physical versus bio-
- Alvatez, M., Fernandez, E., & Ferez, F. F. (1999). All-sea CO₂ nucles in a coastal emodyment anected by upwening. Physical versus bio logical control. Oceanologica Acta, 22(5), 499–515. https://doi.org/10.1016/S0399-1784(00)87683-5
- Baumann, H., & Smith, E. M. (2017). Quantifying metabolically driven pH and oxygen fluctuations in US nearshore habitats at diel to interannual time scales. *Estuaries and Coasts*, 41(4), 1102–1117. Journal article. https://doi.org/10.1007/s12237-017-0321-3
- Borges, A. V., & Abril, G. (2011). Carbon dioxide and methane dynamics in estuaries. In E. Wolanski, & D. S. McLusky (Eds.), Treatise on estuarine and coastal science, volume 5, (pp. 119–161). Waltham: Academic Press. https://doi.org/10.1016/B978-0-12-374711-2.00504-0
- Brodeur, J. R., Chen, B., Su, J., Xu, Y.-Y., Hussain, N., Scaboo, K. M., et al. (2019). Chesapeake Bay inorganic carbon: Spatial distribution and seasonal variability. *Frontiers in Marine Science*, *6*, 99. https://doi.org/10.3389/fmars.2019.00099
- Cai, W.-J. (2011). Estuarine and coastal ocean carbon paradox: CO₂ sinks or sites of terrestrial carbon incineration? *Annual Review of Marine Science*, 3(1), 123–145. https://doi.org/10.1146/annurev-marine-120709-142723
- Cai, W.-J., & Wang, Y. (1998). The chemistry, fluxes, and sources of carbon dioxide in the estuarine waters of the Satilla and Altamaha Rivers, Georgia. *Limnology and Oceanography*, 43(4), 657–668. https://doi.org/10.4319/lo.1998.43.4.0657
- Cai, W.-J., Huang, W.-J., Luther, G. W., Pierrot, D., Li, M., Testa, J., et al. (2017). Redox reactions and weak buffering capacity lead to acidification in the Chesapeake Bay. *Nature Communications*, 8(1), 369. https://doi.org/10.1038/s41467-017-00417-7
- Carstensen, J., Chierici, M., Gustafsson, B. G., & Gustafsson, E. (2018). Long-term and seasonal trends in estuarine and coastal carbonate systems. Global Biogeochemical Cycles, 32, 497–513. https://doi.org/10.1002/2017GB005781
- Carstensen, J., & Duarte, C. M. (2019). Drivers of pH variability in coastal ecosystems. Environmental Science & Technology, 53(8), 4020–4029. https://doi.org/10.1021/acs.est.8b03655
- Chen, C.-T. A., & Borges, A. V. (2009). Reconciling opposing views on carbon cycling in the coastal ocean: Continental shelves as sinks and near-shore ecosystems as sources of atmospheric CO₂. *Deep Sea Research Part II: Topical Studies in Oceanography*, 56(8–10), 578–590. https://doi.org/10.1016/j.dsr2.2009.01.001
- Chen, C.-T. A., Huang, T.-H., Chen, Y.-C., Bai, Y., He, X., & Kang, Y. (2013). Air-sea exchanges of CO₂ in the world's coastal seas. *Biogeosciences*, 10(10), 6509–6544. https://doi.org/10.5194/bg-10-6509-2013

Acknowledgments

This research was supported with funding from the National Science Foundation's Chemical Oceanography Program (OCE-1536096 and OCE-1537013), NASA through Grants NNX14AM37G and NNX14AF93G, and the National Oceanic and Atmospheric Administration's Ocean Acidification Program under Award NA18OAR0170430 to the Virginia Institute of Marine Science. Any use of trade, firm, or product names is for descriptive purposes only and does not imply endorsement by the U.S. Government.



Chen, C.-T. A., Huang, T.-H., Fu, Y.-H., Bai, Y., & He, X. (2012). Strong sources of CO₂ in upper estuaries become sinks of CO₂ in large river plumes. *Current Opinion in Environmental Sustainability*, 4(2), 179–185. http://www.sciencedirect.com/science/article/pii/ S1877343512000188, https://doi.org/10.1016/j.cosust.2012.02.003

Chesapeake Bay Program. (1996). Recommended guidelines for sampling and analyses in the Chesapeake Bay monitoring program. Annapolis, MD: Chesapeake Bay Program.

Chesapeake Bay Program. (2004). Chesapeake Bay program analytical segmentation scheme: Revisions, decisions and rationales, 1983–2003. Annapolis, MD: Chesapeake Bay Program.

Cloern, J. E., Foster, S. Q., & Kleckner, A. E. (2014). Phytoplankton primary production in the world's estuarine-coastal ecosystems. Biogeosciences, 11(9), 2477–2501. https://doi.org/10.5194/bg-11-2477-2014

- Dickson, A. G. (1990). Standard potential of the reaction: AgCl(s) + 1/2H₂(g) = Ag(s) + HCl (aq), and the standard acidity constant of the ion HSO₄⁻ in synthetic sea water from 273.15 to 318.15 K. *The Journal of Chemical Thermodynamics*, 22(2), 113–127. https://doi.org/ 10.1016/0021-9614(90)90074-Z
- Dickson, A. G., & Millero, F. J. (1987). A comparison of the equilibrium constants for the dissociation of carbonic acid in seawater media. Deep Sea Research Part A. Oceanographic Research Papers, 34(10), 1733–1743. https://doi.org/10.1016/0198-0149(87)90021-5

Dinauer, A., & Mucci, A. (2017). Spatial variability in surface-water pCO₂ and gas exchange in the world's largest semi-enclosed estuarine system: St. Lawrence estuary (Canada). *Biogeosciences*, *14*(13), 3221–3237. https://doi.org/10.5194/bg-14-3221-2017

Duarte, C. M., Hendriks, I. E., Moore, T. S., Olsen, Y. S., Steckbauer, A., Ramajo, L., et al. (2013). Is ocean acidification an open-ocean syndrome? Understanding anthropogenic impacts on seawater pH. *Estuaries and Coasts*, 53(8), 4020–4029. https://doi.org/10.1021/acs. est.8b03655

Emerson, S. R., & Hedges, J. H. (2008). Chemical oceanography and the marine carbon cycle. New York, USA: Cambridge University Press. https://doi.org/10.1017/CBO9780511793202

Etcheto, J., & Merlivat, L. (1988). Satellite determination of the carbon dioxide exchange coefficient at the ocean-atmosphere interface: A first step. Journal of Geophysical Research: Oceans, 93(C12), 15,669–15,678. https://doi.org/10.1029/JC093iC12p15669

Feng, Y., Friedrichs, M. A. M., Wilkin, J., Tian, H., Yang, Q., Hofmann, E. E., et al. (2015). Chesapeake Bay nitrogen fluxes derived from a land-estuarine ocean biogeochemical modeling system: Model description, evaluation and nitrogen budgets. *Journal of Geophysical Research: Biogeosciences*, 120, 1666–1695. https://doi.org/10.1002/2015JG002931

Fisher, T. R., Hagy, J. D., & Rochelle-Newall, E. J. (1998). Dissolved and particulate organic carbon in Chesapeake Bay. *Estuaries*, 21(2), 215–229. https://doi.org/10.2307/1352470

Frankignoulle, M., Abril, G. L., Borges, A., Bourge, I., Canon, C., Delille, B., et al. (1998). Carbon dioxide emission from European estuaries. Science, 282(5388), 434–436. https://doi.org/10.1126/science.282.5388.434

Frankignoulle, M., & Borges, A. V. (2001). Direct and indirect pCO₂ measurements in a wide range of pCO₂ and salinity values (the Scheldt estuary). Aquatic Geochemistry, 7(4), 267–273. https://doi.org/10.1023/A:1015251010481

Friedman, J. R., Shadwick, E. H., Friedrichs, M. A. M., Najjar, R. G., De Meo, O. A., Da, F., & Smith, J. L. (2020). Seasonal variability of the CO₂ system in a large coastal plain estuary. *Journal of Geophysical Research: Oceans*, 125, e2019JC015609. https://doi.org/10.1029/ 2019JC015609

Guo, X., Dai, M., Zhai, W., Cai, W. J., & Chen, B. (2009). CO₂ flux and seasonal variability in a large subtropical estuarine system, the Pearl River estuary, China. Journal of Geophysical Research: Biogeosciences, 114, G03013. https://doi.org/10.1029/2008JG000905

Hedges, J., Baldock, J., Gélinas, Y., Lee, C., Peterson, M., & Wakeham, S. (2002). The biochemical and elemental compositions of marine plankton: A NMR perspective. *Marine Chemistry*, 78(1), 47–63. https://doi.org/10.1016/S0304-4203(02)00009-9

Hirsch, R. M., Moyer, D. L., & Archfield, S. A. (2010). Weighted regressions on time, discharge, and season (WRTDS), with an application to Chesapeake Bay river inputs. Journal of the American Water Resources Association, 46(5), 857–880. https://doi.org/10.1111/j.1752-1688.2010.00482.x

Ho, D. T., Coffineau, N., Hickman, B., Chow, N., Koffman, T., & Schlosser, P. (2016). Influence of current velocity and wind speed on air-water gas exchange in a mangrove estuary. *Geophysical Research Letters*, 43, 3813–3821. https://doi.org/10.1002/2016GL068727

Hu, X., Beseres Pollack, J., McCutcheon, M. R., Montagna, P. A., & Ouyang, Z. (2015). Long-term alkalinity decrease and acidification of estuaries in Northwestern Gulf of Mexico. Environmental Science & Technology, 49(6), 3401–3409. https://doi.org/10.1021/es505945p

Hunt, C. W., Salisbury, J. E., & Vandemark, D. (2011). Contribution of non-carbonate anions to total alkalinity and overestimation of pCO₂ in New England and New Brunswick rivers. *Biogeosciences*, 8(10), 3069–3076. https://doi.org/10.5194/bg-8-3069-2011

Jiang, L.-Q., Cai, W.-J., & Wang, Y. (2008). A comparative study of carbon dioxide degassing in river- and marine-dominated estuaries. Limnology and Oceanography, 53(6), 2603–2615. https://doi.org/10.4319/lo.2008.53.6.2603

Kemp, W. M., Boynton, W. R., Adolf, J. E., Boesch, D. F., Boicourt, W. C., Brush, G., et al. (2005). Eutrophication of Chesapeake Bay: Historical trends and ecological interactions. *Marine Ecology Progress Series*, 303, 1–29. https://doi.org/10.3354/meps303001

Kemp, W. M., Smith, E. M., Marvin-DiPasquale, M., & Boynton, W. R. (1997). Organic carbon balance and net ecosystem metabolism in Chesapeake Bay. Marine Ecology Progress Series, 150, 229–248. https://doi.org/10.3354/meps150229

Koné, Y. J.-M., Abril, G., Kouadio, K., Delille, B., & Borges, A. (2009). Seasonal variability of carbon dioxide in the rivers and lagoons of Ivory Coast (West Africa). *Estuaries and Coasts*, 32(2), 246–260. https://doi.org/10.1007/s12237-008-9121-0

Kuliński, K., Schneider, B., Hammer, K., Machulik, U., & Schulz-Bull, D. (2014). The influence of dissolved organic matter on the acid-base system of the Baltic Sea. *Journal of Marine Systems*, 132, 106–115. https://doi.org/10.1016/j.jmarsys.2014.01.011

Laruelle, G. G., Dürr, H. H., Slomp, C. P., & Borges, A. V. (2010). Evaluation of sinks and sources of CO₂ in the global coastal ocean using a spatially-explicit typology of estuaries and continental shelves. *Geophysical Research Letters*, 37, L15607. https://doi.org/10.1029/ 2010GL043691

Malone, T. C., Conley, D. J., Fisher, T. R., Glibert, P. M., Harding, L. W., & Sellner, K. G. (1996). Scales of nutrient-limited phytoplankton productivity in Chesapeake Bay. *Estuaries*, 19(2), 371–385. https://doi.org/10.2307/1352457

Mehrbach, C., Culberson, C. H., Hawley, J. E., & Pytkowicx, R. M. (1973). Measurement of the apparent dissociation constants of carbonic acid in seawater at atmospheric pressure. *Limnology and Oceanography*, *18*(6), 897–907. https://doi.org/10.4319/lo.1973.18.6.0897

Millero, F. J. (1979). The thermodynamics of the carbonate system in seawater. *Geochimica et Cosmochimica Acta*, 43, 1651–1661.

Millero, F. J. (2010). Carbonate constants for estuarine waters. Marine and Freshwater Research, 61(2), 139–142. https://doi.org/10.1071/ MF09254

Millero, F. J., Graham, T. B., Huang, F., Bustos-Serrano, H., & Pierrot, D. (2006). Dissociation constants of carbonic acid in seawater as a function of salinity and temperature. *Marine Chemistry*, 100(1-2), 80–94. https://doi.org/10.1016/j.marchem.2005.12.001

Mukhopadhyay, S., Biswas, H., De, T., Sen, S., & Jana, T. (2002). Seasonal effects on the air-water carbon dioxide exchange in the Hooghly estuary, NE coast of bay of Bengal, India. *Journal of Environmental Monitoring*, 4(4), 549–552. https://doi.org/10.1039/B201614A



- Najjar, R. G., Herrmann, M., Cintrón Del Valle, S. M., Friedman, J., Friedrichs, M. A. M., Goldberger, S., & Shadwick, E. (2020). Alkalinity in tidal tributaries of the Chesapeake Bay. *Journal of Geophysical Research: Oceans, 125*, e2019JC015597. https://doi.org/10.1029/ 2019JC015597
- Patsavas, M. C., Byrne, R. H., Wanninkhof, R., Feely, R. A., & Cai, W.-J. (2015). Internal consistency of marine carbonate system measurements and assessments of aragonite saturation state: Insights from two U.S. coastal cruises. *Marine Chemistry*, 176, 9–20. https://doi. org/10.1016/j.marchem.2015.06.022
- Prasad, M. B. K., Kaushal, S. S., & Murtugudde, R. (2013). Long-term pCO₂ dynamics in rivers in the Chesapeake Bay watershed. Applied Geochemistry, 31(0), 209–215. https://doi.org/10.1016/j.apgeochem.2013.01.006
- Raymond, P. A., Bauer, J. E., & Cole, J. J. (2000). Atmospheric CO₂ evasion, dissolved inorganic carbon production, and net heterotrophy in the York River estuary. *Limnology and Oceanography*, 45(8), 1707–1717. https://doi.org/10.4319/lo.2000.45.8.1707
- Rochelle-Newall, E. J., & Fisher, T. R. (2002). Chromophoric dissolved organic matter and dissolved organic carbon in Chesapeake Bay. *Marine Chemistry*, 77(1), 23–41. https://doi.org/10.1016/S0304-4203(01)00073-1
- Shadwick, E. H., De Meo, O. A., & Friedman, J. R. (2019). Discrete CO₂-system measurements in the Chesapeake Bay mainstem between 2016 and 2018. https://doi.org/10.25773/rntn-ez18

Shadwick, E. H., Friedrichs, M. A. M., Najjar, R. G., De Meo, O. A., Friedman, J. R., Da, F., & Reay, W. G. (2019). High-frequency CO₂ system variability over the winter-to-spring transition in a coastal plain estuary. *Journal of Geophysical Research: Oceans*, 124, 7626–7642. https://doi.org/10.1029/2019JC015246

Shen, C., Testa, J. M., Li, M., Cai, W.-J., Waldbusser, G. G., Ni, W., et al. (2019). Controls on carbonate system dynamics in a coastal plain estuary: A modeling study. *Journal of Geophysical Research: Biogeosciences*, *124*, 61–78. https://doi.org/10.1029/2018JG004802

- Shen, C., Testa, J. M., Ni, W., Cai, W.-J., Li, M., & Kemp, W. M. (2019). Ecosystem metabolism and carbon balance in Chesapeake Bay: A 30-year analysis using a coupled hydrodynamic–biogeochemical model. *Journal of Geophysical Research: Oceans, 124*, 6141–6153. https://doi.org/10.1029/2019JC015296
- Shen, J., & Wang, H. V. (2007). Determining the age of water and long-term transport timescale of the Chesapeake Bay. Estuarine, Coastal and Shelf Science, 74(4), 585–598. https://doi.org/10.1016/j.ecss.2007.05.017
- Stets, E. G., & Striegl, R. G. (2012). Carbon export by rivers draining the conterminous United States. Inland Waters, 2(4), 177–184. https://doi.org/10.5268/IW-2.4.510
- St-Laurent, P., Friedrichs, M. A. M., Najjar, R. G., Shadwick, E. H., Tian, H., Yao, Y., & Stets, E. G. (2020). Relative impacts of global changes and regional watershed changes on the inorganic carbon balance of the Chesapeake Bay. *Biogeosciences Discussions, in review.*. https:// doi.org/10.5194/bg-2020-117
- U.S. Geological Survey [USGS] (2019). Data inventory page for site 01578310—Susquehanna River at Conowingo, Maryland: U.S. Geological Survey web page. accessed July 20, 2019. at https://waterdata.usgs.gov/nwis/inventory/?site no=14202980

Uppström, L. R. (1974). The boron/chlorinity ratio of deep-sea water from the Pacific Ocean. *Deep Sea Research*, 21(2), 161–162. https://doi. org/10.1016/0011-7471(74)90074-6

- van Heuven, S., Pierrot, D., Rae, J. W. B., Lewis, E., & Wallace, D. W. R. (2011). MATLAB program developed for CO₂ system calculations, ORNL/CDIAC-105b. Oak Ridge, Tennessee: Carbon Dioxide Information Analysis Center, Oak Ridge National Laboratory, U.S. Department of Energy.
- Waldbusser, G. G., Powell, E. N., & Mann, R. (2013). Ecosystem effects of shell aggregations and cycling in coastal waters: An example of Chesapeake Bay oyster reefs. Ecology, 94(4), 895–903. https://doi.org/10.1890/12-1179.1

Wanninkhof, R. (2014). Relationship between wind speed and gas exchange over the ocean revisited. Limnology and Oceanography: Methods, 12(6), 351–362. https://doi.org/10.4319/lom.2014.12.351

- Yang, B., Byrne, R. H., & Lindemuth, M. (2015). Contributions of organic alkalinity to total alkalinity in coastal waters: A spectrophotometric approach. Marine Chemistry, 176, 199–207. https://doi.org/10.1016/j.marchem.2015.09.008
- Zhai, W., Dai, M., & Guo, X. (2007). Carbonate system and CO₂ degassing fluxes in the inner estuary of Changjiang (Yangtze) river, China. Marine Chemistry, 107(3), 342–356. https://doi.org/10.1016/j.marchem.2007.02.011
- Zhang, Q., Murphy, R. R., Tian, R., Forsyth, M. K., Trentacoste, E. M., Keisman, J., & Tango, P. J. (2018). Chesapeake Bay's water quality condition has been recovering: Insights from a multimetric indicator assessment of thirty years of tidal monitoring data. *Science of the Total Environment*, 637-638, 1617–1625. https://doi.org/10.1016/j.scitotenv.2018.05.025